

OPEN ACCESS

Citation: Coscione S, Esau T, Kekeubata E, Diau J, Asugeni R, MacLaren D, et al. (2018) Impact of ivermectin administered for scabies treatment on the prevalence of head lice in Atoifi, Solomon Islands. PLoS Negl Trop Dis 12(9): e0006825. https://doi.org/10.1371/journal.pntd.0006825

Editor: Michele Murdoch, Watford General Hospital, UNITED KINGDOM

Received: June 13, 2018

Accepted: September 7, 2018

Published: September 25, 2018

Copyright: © 2018 Coscione et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data are in the supporting information.

Funding: MM was supported by a Wellcome Trust Fellowship (102807). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Impact of ivermectin administered for scabies treatment on the prevalence of head lice in Atoifi, Solomon Islands

Suny Coscione¹, Tommy Esau², Esau Kekeubata², Jason Diau², Rowena Asugeni², David MacLaren³, Andrew C. Steer^{4,5}, Christian Kositz^{1‡}, Michael Marks^{1,6‡}*

 Clinical Research Department, Faculty of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Atoifi Adventist Hospital, Atoifi, Malaita, Solomon Islands,
College of Medicine and Dentistry, James Cook University, Cairns, Queensland, Australia, 4 Centre for International Child Health, University of Melbourne, Department of Paediatrics, Melbourne, Australia,
Department of General Medicine, Royal Children's Hospital, Melbourne, Australia, 6 Hospital for Tropical Diseases, University College London Hospitals NHS Trust, London, United Kingdom

‡ These authors are joint senior authors on this work.

* michael.marks@lshtm.ac.uk

Abstract

Background

Scabies and head lice are ubiquitous ectoparasitic infestations that are common across the Pacific Islands. Ivermectin is an effective treatment for both conditions, although the doses used vary. At a community level, mass drug administration (MDA) with ivermectin is an effective strategy to decrease prevalence of scabies. To what extent MDA with ivermectin will also reduce prevalence of head lice is unknown.

Methodology

Head lice prevalence was assessed before and after MDA with oral ivermectin (at a dose of 200 micrograms per kilogram of body weight) administered on day 1 and day 8. The primary outcome was the change in prevalence of head louse infestation at two weeks compared to baseline. Longer term efficacy was assessed three months after MDA.

Results

118 participants were enrolled. Baseline prevalence of active head louse infestation was 25.4% (95% CI 18.4–34.0). At two-week follow-up, prevalence was 2.5% (95% CI 0.9–7.2), a relative reduction of 89.1% (95% CI 72.7–91.4%, p <0.001). At three-month follow-up, prevalence was 7.5% (95% CI 2.7–12.3), a relative reduction of 70.6% (95% CI 72.7%-91.4%, p <0.001). Head louse infestation was associated with younger age (age \leq 10 years: prevalence 46.7%; adjusted odds ratio compared to adults of 7.2, 95% CI 2.0–25.9) and with having at least one other member of the household with active head louse infestation (adjusted odds ratio 4.3, 95% CI 1.7–11.1).

Conclusions

Head louse infestation is common in the Solomon Islands. This proof of principle study shows that oral ivermectin at a dose of 200 micrograms per kilogram can reduce the burden

of active head louse infestation, offering an additional collateral benefit of MDA with ivermectin for scabies control.

Trial registration

ClinicalTrials.gov NCT03236168.

Author summary

Head lice and scabies are both caused by ectoparasites and lead to itchy skin conditions that are associated with secondary bacterial infections and social stigma. Both are common in developing countries. In the Solomon Islands, mass treatment of communities using ivermectin (at dose of 200 micrograms per kilogram) has been shown to be an effective strategy in reducing scabies prevalence. This study shows that the same dose of ivermectin and the same regime of mass drug administration is also effective at reducing the burden of head louse infestation, offering an additional benefit to community wide treatment.

Introduction

The human head louse, *Pediculus humanus capitis*, is an ectoparasitic insect found on the scalp that is endemic world-wide [1]. Head louse infestation can cause intense scalp itching and scratching can lead to secondary bacterial infections [2]. In low and middle income countries, this parasitic skin disease has received little attention in research [3]. Specifically, in the Pacific region, there is minimal research quantifying its prevalence or potential treatment strategies [4].

Ivermectin is an oral, semi-synthetic derivative of the avermectin family of lactones that selectively binds glutamate-gated chloride channels found in invertebrate muscle and nerve cells thus disrupting neurotransmission in a wide range of human parasites [5]. Oral preparations of ivermectin can be used to treat head lice [2,6] as well as other blood feeding ectoparasitic diseases, including scabies [7,8]. It should be noted that for both, head lice and scabies, oral ivermectin treatment does not have ovicidal action.

A large trial evaluating the use of ivermectin monotherapy, in difficult-to-treat head louse infestation, showed that ivermectin at a dose of 400 micrograms per kilogram (kg) on day 1 and day 8, had superior efficacy compared to 0.5% malathion lotion [5]. Small studies have suggested variable efficacy when a dose of 200 micrograms per kg is used. Single administration resulted in variable reductions in head lice of 45% - 100% at one week follow up but repeat administration on day 8 yielded higher reductions (87.5% - 100%) at two-week follow up [9–13]. These studies have all focused on treatment of individual patients and not on whole communities where head lice are common and where re-infestation from other community members might be anticipated to result in a lower efficacy than that seen in individual treatment trials.

Scabies, another ectoparasitic disease, has already been shown to be extremely common in the Pacific [14–16]. Ivermectin, used as part of mass drug administration (MDA), at a dose of 200 micrograms per kg given at a 7-day interval, has been shown to be an effective strategy to reduce community wide prevalence of scabies in the region [17–20]. These programmes might have ancillary benefits such as treatment of headlice in the communities.

The study we conducted aimed to establish baseline prevalence of head louse infestation in a rural community in the Solomon Islands and to assess whether MDA using ivermectin, at the lower dose of 200 micrograms per kilogram (the same regimen already used for scabies treatment) would be an effective method to lower community prevalence of head lice.

Methods

Ethics statement

The study was approved by the London School of Hygiene and Tropical Medicine (LSHTM) ethics committee and the National Health Research and Ethics Committee of the Solomon Islands Ministry of Health and Medical Services. Written consent was obtained from all participants by a staff member fluent in both local languages (Pijin and East Kwaio). Parents or guardians provided written consent for participants aged below 18 years. Children were additionally asked to provide verbal assent for participation. The study was prospectively registered on clinicaltrials.gov (NCT03236168).

Study Site

The study was conducted in the campus area of Atoifi Adventist Hospital (AAH) in the East Kwaio region of Malaita Island, Solomon Islands. The hospital staff and their families live on campus and create a self-contained community of approximately 200 individuals. The campus has a school for children aged up to 16 years. Students who attend Atoifi School live on campus (staff children) or live in villages that surround the campus. All individuals living on AAH campus and all children and families of children attending Atoifi School were eligible for the study.

Data collection and ivermectin administration

At baseline participants' demographic and household data were collected and all individuals underwent a standardised examination of their skin and scalp.

Scalp examination consisted of direct visual inspection of sites of predilection for head lice: the back of the ears, temples and neck. This was followed by parting the hair into 4 sections (first down sagittal plane and then ear to ear). Each section was visually inspected and examined using a metal nit comb with plastic base and 0.1mm teeth size by the brand InfectoPedicul manufactured by InfectoPharm. Eggs and lice were as far as possible not deliberately removed from the hair during the examination to avoid biasing outcome measure assessment by physical delousing. Scalp examinations were carried out by trained nursing staff at Atoifi Hospital led by a physician (SC). Prior to study commencement all staff received a standardised half-day training workshop including practical sessions on clinical examination and use of nit combs. Ten percent of hair examinations were repeated by an independent examiner to ascertain accuracy of diagnosis.

Adult and nymphal stages of head lice were classified as 'active head louse infestation'. If only head lice eggs were seen, this was recorded as 'egg infestation'. No attempts were made to differentiate between viable and non-viable eggs.

The diagnosis of scabies was made clinically based on presence of pruritic inflammatory papules or nodules with a typical anatomical distribution using previously validated criteria [21]. The distribution and number of scabies lesions were recorded and used to classify severity of scabies based on classifications used in previous studies in the Pacific region [16,22]. Participants were not re-examined for scabies as part of this study.

Ivermectin was administered as directly observed therapy at a dose of 200 micrograms per kg of body weight. The first administration of ivermectin was at time of baseline examination. A second administration of ivermectin was given 7 days later to all study participants (day 8).

Participants with contraindications to ivermectin treatment (weight below 12.5kg, pregnancy or breastfeeding) were offered malathion 0.5% lotion for head lice and permethrin 5% cream for scabies. Participants diagnosed with any non-scabies skin condition were briefly counselled and, if necessary, advised to attend the medical clinic to obtain treatment as per standard local protocol.

To assess immediate efficacy, a subgroup of participants was re-examined at 48 hours. School children were chosen both for convenience sampling and because they were predicted (from risk factors identified in literature citation) to have highest rates of head louse infestation. The primary outcome of the study was change in prevalence of head lice at 2 weeks compared to baseline. Longer-term efficacy was assessed by examination at 3 months.

Statistical analysis

For analysis, age was classified into three categories: less than or equal to 10 years, 11–20 years and over or equal to 21 years of age. Area of residence was classified into two categories: household in AAH or household out of AAH campus.

We calculated the prevalence of individuals with adult or nymphal stages of head lice on examination (active head lice) and the prevalence of head lice eggs on examination.

Univariable logistic regression was used to identify risk factors for the presence of head lice at baseline. Variables associated in the univariable regression analysis were included in a multivariable model. We considered age and gender as forced confounders to be included in the multivariable model.

Treatment efficacy was assessed by comparing the proportion of individuals with head lice, using a two-sample test of proportion at baseline compared to 48 hours, two weeks and three months respectively. We calculated the absolute and relative reduction in the prevalence of both active head louse infestation and egg infestation. Data was analysed with STATA software version 14.2 (Stata Corporation, College Station, TX, USA).

Results

There were 218 individuals in the study community. Of these, 118 participants consented to take part in the study (Fig 1). Five participants had contraindications to ivermectin (2 women were breast feeding and 3 children weighed less than 12.5kg) and were given malathion lotion and permethrin cream as per protocol.

The median age in the study population was 20 years (interquartile range 10–26). Thirtyseven individuals (31.4% of cohort) lived in one of the three villages closest to AAH campus (Table 1). Households outside of the AAH campus were on average larger and had a larger number of children (3.4) compared to households on the AAH campus (2.4) (p = 0.001).

Baseline prevalence of head lice and scabies

The baseline prevalence of active head louse infestation was 25.4% (95% CI 18.4–34.0). In all cases where active head louse infestation was identified, head lice eggs were also seen. The baseline prevalence of head lice eggs (with or without active head louse infestation) was 42.3% (95% CI 30.8–48.8). Scabies prevalence was 10.2% (95% CI 5.9–16.9) (Table 2). No cases of crusted scabies were identified. Other skin conditions diagnosed included tinea corporis (n = 3) and eczema (n = 4).





https://doi.org/10.1371/journal.pntd.0006825.g001

Table 1. Demographic characteristics of study cohort.

Study cohort (n = 118)
57 (48.3%)
61 (51.7%)
32 (27.1%)
31 (26.2%)
55 (46.6%)
81 (68.6%)
37 (31.4%)

https://doi.org/10.1371/journal.pntd.0006825.t001

	Active head louse infestation	Head lice eggs	Scabies
Categories	Cases/total in each category	Cases/total in each category	Cases/total in each category
Total	30 /118 (25.4%)	50/118 (42.3%)	12/118 (10.2%)
Age (years)			
≤ 10	14/32	21/32	8/32
11-20	12/31	17/31	1/31
≥21	4/55	12/55	3/55
Gender			
Male	12/61	17/61	4/61
Female	18/57	33/57	8/57

Table 2. Baseline prevalence of head lice and scabies.

https://doi.org/10.1371/journal.pntd.0006825.t002

Risk factors for head lice

Active head louse infestation was associated with younger age, with increased risk compared to adults for both 0–10 years (AOR 7.2, 95% CI 2.0–25.9, p = 0.003) and 11–20 years (AOR 9.1, 95% CI 2.4–34.2, p = 0.001). Infestation was also associated with the presence of other members of the household with active lice infestation (AOR 3.9, 95% CI 1.1–14.3, p = 0.040). No association was seen with gender or location of the house on or off the AAH campus (Table 3).

The presence of head lice eggs was associated with age, with highest risk in children aged under or equal to 10 years (AOR 5.9, 95% CI 2.0–17.1). Head lice eggs were also associated with female gender (AOR 3.1, 95% CI 1.3–7.6) (Table 3).

Change in head lice prevalence at 48 hours, 2 weeks and 3 months

Twenty-four out of 28 school-children (85.7%) were re-examined at 48 hours following ivermectin administration. The prevalence of active infestation decreased from 45.8% at baseline to 0%.

At two weeks after MDA, the prevalence of active infestation had declined significantly (25.6% vs 2.5%, relative reduction (RR) 89.1%, p < 0.001) (Table 4).

At three months, the prevalence of active head louse infestation remained significantly lower than at baseline (25.6% vs 7.5%, RR 70.6%, p < 0.001) (Table 4).

The prevalence of head lice eggs was unchanged at two weeks (42.3% vs 41.8%) but reduced significantly at three months (42.3% vs 19.6%, RR 53.7%, p < 0.001) (Table 4).

Discussion

This is the first study to demonstrate the efficacy of community-based ivermectin administration for head lice treatment using ivermectin 200 micrograms per kg, administered twice at day 1 and day 8. This regimen significantly reduced prevalence of active head louse infestation at both the two-week and three-month follow-up visit.

At 48 hours, none of the subgroup screened had active head louse infestation. This suggests that the 200 micrograms per kg dose is effective at killing nymphal and adult head lice stages in this population and that a higher dose (400 microgram per kg) is unlikely to offer any additional short-term head lice killing benefits.

At two weeks, the prevalence of active head lice in the whole study sample had decreased significantly (from 25.6% to 2.5%). There was no reduction in head lice eggs at two weeks, however, this was expected given that head lice eggs are cemented, via glue-like glandular discharge, to the hair shaft and remain attached even after hatching. Therefore, head lice eggs will still be found even after the active head louse infestation has been treated.

	Active head louse infestation		Head lice eggs	
	Adjusted Odds ratio (95% CI)	p-value	Adjusted Odds ratio (95% CI)	p-value
Age (y)				
≤ 10	7.2 (2.0–25.9)	0.003	5.9 (2.0–17.1)	0.001
11–20	9.1 (2.4–34.2)	0.001	5.2 (1.8–15.0)	0.002
\geq 21	1.0	-	1.0	-
Gender				
Male	1.0	-	1.0	-
Female	1.5 (0.6–4.1)	0.377	3.1 (1.3–7.6)	0.010
Location of house				
AAH campus	1.0	-	1.0	-
Villages	1.0 (0.3–3.2)	0.956	1.5 (0.5-4.9)	0.491
Others infested in household				
None	1.0		1.0	
At least one person with active head louse infestation	3.9 (1.1–14.3)	0.040	2.2 (0.7–6.7)	0.179

Table 3. Multivariate analysis of risk factors for head lice.

https://doi.org/10.1371/journal.pntd.0006825.t003

At three months, the effect of the intervention was sustained. The prevalence of active infestation remained significantly lower than baseline. Furthermore, the reduction in active head louse infestation translated into a reduction in egg prevalence at three months.

Our study also demonstrates that head lice are an extremely common ectoparasitic infection in the Solomon Islands with prevalence comparable or higher than those reported in other lower income settings such as Nepal, Brazil and Egypt [3]. This is a significant burden of disease and further qualitative research is important to elucidate the local attitudes and perception towards head louse infestation.

The findings of this study highlight an opportunity to concomitantly target two common ectoparasitic infections at community level with a single intervention strategy. Showing evidence for additional benefits of MDA may increase engagement of local communities and support from local authorities towards community-wide programs and enables control of head louse infestation to be integrated with growing control programs for neglected tropical diseases such as scabies.

Our study had some limitations. First, MDA coverage was 54% of the eligible population. We did not record the age of non-participants and, as the prevalence of headlice is higher in children, this may have biased our overall prevalence estimate. Despite this our intervention demonstrated efficacy of MDA. Second, prevalence of active head louse infestation did increase slightly at three months compared to two weeks. This may well reflect re-introduction of head lice from untreated members of the community or surrounding villages. Expanding the population that receives treatment may reduce the risk of re-introduction and thus

Head louse infestation	Baseline prevalence	2-week prevalence	3-month prevalence	Absolute Reduction in Prevalence (baseline to 3 months)	Relative reduction in prevalence (baseline to 3 months)				
	Cases/n (%)	Cases/n (%)	Cases/n (%)	Percentage points (95% CI)	Percentage points (95% CI)	P value			
Active	30/118 (25.6)	3/117 (2.5)	8/107 (7.5)	18.12 (8.6–27.2)	70.6 (72.7–91.4)	< 0.001			
Eggs	50/118 (42.3)	49/117 (41.8)	21/107 (19.6)	22.7 (11.1–34.4)	53.7 (28.3–70.1)	< 0.001			

Table 4. Change in prevalence of head louse infestation over 3 months.

https://doi.org/10.1371/journal.pntd.0006825.t004

improve long-term reduction in head lice prevalence at community level. Finally, we did not differentiate between viable eggs and non-viable eggs. Being able to reliably distinguish viability of eggs would have yielded a more accurate estimate of active head louse infestation in the community and could be considered in future studies.

Our study demonstrated the effectiveness of a two-stage administration of ivermectin at a dose of 200microgram per kg for the treatment of head lice. A reduction in head lice prevalence is likely to be an ancillary benefit of the scale-up of scabies control programmes in the Pacific region and elsewhere. As with other ectoparasitic infestations, head lice are a common and under-recognised parasitic infestation in the Solomon Islands. Fully measuring the impact of MDA programs can maximise the potential benefits to the community and aid community engagement with the intervention. Further delineating the full range of benefits of ivermectin MDA should be a priority for scabies control programmes.

Supporting information

S1 Checklist. STROBE Checklist. (DOCX)

S1 File. Study data. (XLSX)

Author Contributions

Conceptualization: Christian Kositz, Michael Marks.

Data curation: Suny Coscione, Michael Marks.

- Formal analysis: Suny Coscione, David MacLaren, Andrew C. Steer, Christian Kositz, Michael Marks.
- Funding acquisition: Michael Marks.
- **Investigation:** Suny Coscione, Tommy Esau, Esau Kekeubata, Christian Kositz, Michael Marks.
- Methodology: Suny Coscione, Tommy Esau, Esau Kekeubata, Jason Diau, Rowena Asugeni, David MacLaren, Andrew C. Steer, Christian Kositz, Michael Marks.
- **Project administration:** Tommy Esau, Esau Kekeubata, Jason Diau, Rowena Asugeni, David MacLaren, Christian Kositz, Michael Marks.

Resources: Jason Diau, Rowena Asugeni, Michael Marks.

Supervision: Michael Marks.

Writing - original draft: Suny Coscione.

Writing – review & editing: Tommy Esau, Esau Kekeubata, Jason Diau, Rowena Asugeni, David MacLaren, Andrew C. Steer, Christian Kositz, Michael Marks.

References

- 1. Centre for Disease Control & Prevention (CDC). CDC—Head Lice—General Information. In: CDC [Internet]. [cited 30 Aug 2017]. Available: https://www.cdc.gov/parasites/lice/head/gen_info/faqs.html
- Van der Wouden JC, Klootwijk T, Le Cleach L, Do G, Vander Stichele R, Knuistingh Neven A, et al. Interventions for treating head lice. Cochrane Database. Wiley Online Library; 2011; https://doi.org/10.1093/database/baq027

- Falagas ME, Matthaiou DK, Rafailidis PI, Panos G, Pappas G. Worldwide prevalence of head lice. Emerg Infect Dis. Centers for Disease Control and Prevention; 2008; 14: 1493–4. https://doi.org/10. 3201/eid1409.080368 PMID: 18760032
- 4. Speare R, Harrington H, Canyon D, Massey PD. A systematic literature review of pediculosis due to head lice in the Pacific Island Countries and Territories: what country specific research on head lice is needed? BMC Dermatol. 2014; 14: 11. https://doi.org/10.1186/1471-5945-14-11 PMID: 24962507
- Chosidow O, Giraudeau B, Cottrell J, Izri A, Hofmann R, Mann SG, et al. Oral Ivermectin versus Malathion Lotion for Difficult-to-Treat Head Lice. N Engl J Med. Massachusetts Medical Society; 2010; 362: 896–905. https://doi.org/10.1056/NEJMoa0905471 PMID: 20220184
- Sanchezruiz WL, Nuzum DS, Kouzi SA. Oral ivermectin for the treatment of head lice infestation. Am J Health-Syst Pharm AJHP Off J Am Soc Health-Syst Pharm. American Society of Health-System Pharmacists; 2018; ajhp170464. https://doi.org/10.2146/ajhp170464 PMID: 29789316
- Usha V, Gopalakrishnan Nair T. A comparative study of oral ivermectin and topical permethrin cream in the treatment of scabies☆, ☆☆. J Am Acad Dermatol.: Merck & Co, West Point (PA); 2000; 42: 236– 240. https://doi.org/10.1016/S0190-9622(00)90131-2 PMID: 10642678
- Strong M, Johnstone PW. Interventions for treating scabies. Cochrane Database of Systematic Reviews. Wiley Online Library; 2007. https://doi.org/10.1002/14651858.CD000320.pub2 PMID: 17636630
- Ahmad HM, Abdel-Azim ES, Abdel-Aziz RT. Assessment of topical versus oral ivermectin as a treatment for head lice. Dermatol Ther. Wiley/Blackwell (10.1111); 2014; 27: 307–310. https://doi.org/10. 1111/dth.12144 PMID: 25041547
- Nofal A. Oral ivermectin for head lice: a comparison with 0.5% topical malathion lotion. JDDG J Dtsch Dermatol Ges. 2010; 8: 985–988. https://doi.org/10.1111/j.1610-0387.2010.07487.x PMID: 20718901
- Ameen M, Arenas R, Villanueva-Reyes J, Ruiz-Esmenjaud J, Millar D, Domínguez-Dueñas F, et al. Oral ivermectin for treatment of pediculosis capitis. Pediatr Infect Dis J. 2010; 29: 991–3. PMID: 21046698
- Currie MJ, Reynolds GJ, Glasgow NJ, Bowden FJ. A Pilot Study of the Use of Oral Ivermectin to Treat Head Lice in Primary School Students in Australia. Pediatr Dermatol. 2010; 27: 595–599. https://doi. org/10.1111/j.1525-1470.2010.01317.x PMID: 21138467
- 13. Glaziou P, Nyguyen LN, Moulia-Pelat JP, Cartel JL, Martin PM. Efficacy of ivermectin for the treatment of head lice (Pediculosis capitis). Trop Med Parasitol Off Organ Dtsch Tropenmedizinische Ges Dtsch Ges Tech Zusammenarbeit GTZ. 1994; 45: 253–4.
- Steer AC, Jenney AWJ, Kado J, Batzloff MR, La Vincente S, Waqatakirewa L, et al. High Burden of Impetigo and Scabies in a Tropical Country. Franco-Paredes C, editor. PLoS Negl Trop Dis. Fiji Ministry of Health; 2009; 3: e467. https://doi.org/10.1371/journal.pntd.0000467 PMID: 19547749
- Romani L, Whitfeld M, Koroivueta J, Kama M, Wand H, Tikoduadua L, et al. The Epidemiology of Scabies and Impetigo in Relation to Demographic and Residential Characteristics: Baseline Findings from the Skin Health Intervention Fiji Trial. Am J Trop Med Hyg. 2017; 97: 845–850. <u>https://doi.org/10.4269/</u> ajtmh.16-0753 PMID: 28722612
- Mason DS, Marks M, Sokana O, Solomon AW, Mabey DC, Romani L, et al. The Prevalence of Scabies and Impetigo in the Solomon Islands: A Population-Based Survey. Vinetz JM, editor. PLoS Negl Trop Dis. World Health Organisation, Department of Child and Adolescent Health and Development; 2016; 10: e0004803. https://doi.org/10.1371/journal.pntd.0004803 PMID: 27348119
- Haar K, Romani L, Filimone R, Kishore K, Tuicakau M, Koroivueta J, et al. Scabies community prevalence and mass drug administration in two Fijian villages. Int J Dermatol. 2014; 53: 739–745. https://doi. org/10.1111/ijd.12353 PMID: 24168177
- Romani L, Whitfeld MJ, Koroivueta J, Kama M, Wand H, Tikoduadua L, et al. Mass Drug Administration for Scabies Control in a Population with Endemic Disease. N Engl J Med. 2015; 373: 2305–2313. https://doi.org/10.1056/NEJMoa1500987 PMID: 26650152
- Marks M, Taotao-Wini B, Satorara L, Engelman D, Nasi T, Mabey DC, et al. Long Term Control of Scabies Fifteen Years after an Intensive Treatment Programme. Vinetz JM, editor. PLoS Negl Trop Dis. John Wiley & Sons, Ltd; 2015; 9: e0004246. https://doi.org/10.1371/journal.pntd.0004246 PMID: 26624616
- 20. Lawrence G, Leafasia J, Sheridan J, Hills S, Wate J, Wate C, et al. Control of scabies, skin sores and haematuria in children in the Solomon Islands: another role for ivermectin. Bull World Health Organ. Switzerland; 2005; 83: 34–42. https://doi.org//S0042-96862005000100012 PMID: 15682247
- Steer A, Tikoduadua L, M Manalac E, Colquhoun S, Carapetisd J, Maclennane C. Validation of an Integrated Management of Childhood Illness algorithm for managing common skin conditions in Fiji. Bull World Health Organ. 2009; 87: 173–179. https://doi.org/10.2471/BLT.08.052712 PMID: 19377712

- 22. Hay RJ, Steer AC, Engelman D, Walton S, Carapetis J, Currie B. Scabies in the developing world-its prevalence, complications, and management. Clin Microbiol Infect Off Publ Eur Soc Clin Microbiol Infect Dis. World Bank and Oxford University Press Publications, New York; 2012; 18: 313–23. https://doi.org/10.1111/j.1469-0691.2012.03798.x PMID: 22429456
- Consolidated Standards of Reporting Trials (CONSORT). CONSORT 2010 Key Documents [Internet]. 2010 [cited 29 Aug 2017]. Available: http://www.consort-statement.org/